Author’s response to reviews

Title: "Happy Feet": Evaluating the benefits of a 100-day 10,000 step challenge on mental health and wellbeing

Authors:

Karen Hallam (Karen.Hallam@vu.edu.au)

Shane Bilsborough (shane.bilsborough@stepathlon.com)

Maximilian Pangratius de Courten (maximilian.decourten@vu.edu.au)

Version: 2 Date: 13 Jan 2018

Author’s response to reviews:

Our actions on the recent minor revision requirements: 1. Under your Ethics approval heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. - A clarifying sentence for the written consent obtained was added; 2. Please include the email addresses of all authors on the title page of your manuscript. - email addresses inserted; 3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list. - list of abbreviations added; and in-text definition added at first use; 4. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. - manuscript edited for English language; 5. Please remove the cover letter and the response to reviewers from the file inventory, as they are no longer needed at this stage of the editorial process. - will be removed; 6. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process. - clean copy generated to be attached.